

Answering Reviewers:

Reviewer #1:

1. The cause of death in patients with recurrent HCC were added to manuscript.
2. The criteria to select a patient for open versus laparoscopic liver resection were previously described in reference no. 30.

Reviewer #2

1. Abstract and Conclusions revised as suggested